Author's response to reviews

Title: Anesthesiologists' practice patterns for treatment of postoperative nausea and vomiting in the ambulatory Post Anesthesia Care Unit

Authors:

Alex Macario (amaca@stanford.edu)
Louis Claybon (leclaybon@hotmail.com)
Joseph V Pergolizzi (jpjmd@msn.com)

Version: 3 Date: 22 May 2006

Author's response to reviews:

- Replace PACU with Post-Anesthesia Care Unit (PACU) within the title (in both the manuscript file and submission system).
  DONE

- Please give more detail within the background section of the abstract to give the context and purpose of the study.
  DONE

- Please acknowledge anyone who contributed towards the study by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include their source(s) of funding. Please also acknowledge anyone who contributed materials essential for the study.
  NO ONE OTHER THAN AUTHORS

Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements.
  NOT APPLICABLE

Please list the source(s) of funding for the study, for each author, and for the manuscript preparation in the acknowledgements section. Authors must describe the role of the funding body, if any, in study design; in the collection, analysis, and interpretation of data; in the writing of the manuscript; and in the decision to submit the manuscript for publication.
  DONE

We would also request that you go though the manuscript formatting checklist one more time and ensure that your revised manuscript conforms to all of the points. The link to the formatting checklist is provided at the bottom of this e-mail.

When you have revised your manuscript in light of the reviewers' comment and made any necessary changes to the format, please upload the revised version by following the instructions provided below.

DONE